Author's response to reviews

Title: Complete resection of a giant mediastinal teratoma occupying the entire right hemithorax in a 14-year-old boy

Authors:

Honglin Zhao (zhaohonglintj@163.com)
Daxing Zhu (zhudaxingtj@163.com)
Qinghua Zhou (cnliuqian@gmail.com)

Version: 3  Date: 11 June 2014

Author's response to reviews:

Dear Editor,

Thanks for your comments. We've revised the article according to the suggestions.

Review 1:1. This is a well done article.
2. The authors are to be congratulated on the care of their patient.
Answer: Thanks for your comments.

Reviewer 2: The case is interesting. I could only raise one point that the references should be updated due to some publications related to your work in recent years.
Answer 2: Thank you. We’ve revised some sentences in the article.

sincerely
Qinghua Zhou M.D.